UNIVERSITY of York

This is a repository copy of *Microbes increase thermal sensitivity in the mosquito Aedes aegypti, with the potential to change disease distributions.*

White Rose Research Online URL for this paper: <u>https://eprints.whiterose.ac.uk/188785/</u>

Version: Published Version

Article:

Ware-Gilmore, Fhallon, Sgrò, Carla M, Xi, Zhiyong et al. (6 more authors) (2021) Microbes increase thermal sensitivity in the mosquito Aedes aegypti, with the potential to change disease distributions. PLOS NEGLECTED TROPICAL DISEASES. e0009548. ISSN 1935-2735

https://doi.org/10.1371/journal.pntd.0009548

Reuse

This article is distributed under the terms of the Creative Commons Attribution (CC BY) licence. This licence allows you to distribute, remix, tweak, and build upon the work, even commercially, as long as you credit the authors for the original work. More information and the full terms of the licence here: https://creativecommons.org/licenses/

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



eprints@whiterose.ac.uk https://eprints.whiterose.ac.uk/ RESEARCH ARTICLE

Microbes increase thermal sensitivity in the mosquito *Aedes aegypti*, with the potential to change disease distributions

Fhallon Ware-Gilmore¹, Carla M. Sgrò², Zhiyong Xi³, Heverton L. C. Dutra¹, Matthew J. Jones¹, Katriona Shea⁴, Matthew D. Hall², Matthew B. Thomas¹, Elizabeth A. McGraw^{1,4}*

1 Department of Entomology & The Center for Infectious Disease Dynamics, The Pennsylvania State University, University Park, Pennsylvania, United States of America, 2 School of Biological Sciences, Monash University, Melbourne, Victoria, Australia, 3 Department of Microbiology and Molecular Genetics, Michigan State University, East Lansing, Michigan, United States of America, 4 Department of Biology & The Center for Infectious Disease Dynamics, The Pennsylvania State University, University Park, Pennsylvania, United States of America

* eam7@psu.edu

Abstract

The mosquito Aedes aegypti is the primary vector of many disease-causing viruses, including dengue (DENV), Zika, chikungunya, and yellow fever. As consequences of climate change, we expect an increase in both global mean temperatures and extreme climatic events. When temperatures fluctuate, mosquito vectors will be increasingly exposed to temperatures beyond their upper thermal limits. Here, we examine how DENV infection alters Ae. aegypti thermotolerance by using a high-throughput physiological 'knockdown' assay modeled on studies in Drosophila. Such laboratory measures of thermal tolerance have previously been shown to accurately predict an insect's distribution in the field. We show that DENV infection increases thermal sensitivity, an effect that may ultimately limit the geographic range of the virus. We also show that the endosymbiotic bacterium Wolbachia pipientis, which is currently being released globally as a biological control agent, has a similar impact on thermal sensitivity in Ae. aegypti. Surprisingly, in the coinfected state, Wolbachia did not provide protection against DENV-associated effects on thermal tolerance, nor were the effects of the two infections additive. The latter suggests that the microbes may act by similar means, potentially through activation of shared immune pathways or energetic tradeoffs. Models predicting future ranges of both virus transmission and Wolbachia's efficacy following field release may wish to consider the effects these microbes have on host survival.

Author summary

Changes in global climate, which include higher temperatures and more frequent extreme temperature events, are expected to cause dramatic shifts in the distributions of infectious diseases. The geographic range of the mosquito *Aedes aegypti* continues to expand, risking



G OPEN ACCESS

Citation: Ware-Gilmore F, Sgrò CM, Xi Z, Dutra HLC, Jones MJ, Shea K, et al. (2021) Microbes increase thermal sensitivity in the mosquito *Aedes aegypti*, with the potential to change disease distributions. PLoS Negl Trop Dis 15(7): e0009548. https://doi.org/10.1371/journal.pntd.0009548

Editor: Joshua B. Benoit, University of Cincinnati, UNITED STATES

Received: February 19, 2021

Accepted: June 10, 2021

Published: July 22, 2021

Copyright: © 2021 Ware-Gilmore et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data are available from figshare at DOI <u>10.6084/m9.figshare</u>. 12943454.

Funding: This work was supported, in part, by the International Networks of Excellence scheme between Penn State University and Monash University awarded to CMS, MDH, MBT and EAM. Additionally, the work was supported by Startup funds from Penn State to EAM and a Bunton Waller Fellowship awarded to FWG. The funders had no role in study design, data collection and analysis,

decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

greater incidence of viral diseases including dengue (DENV), Zika, chikungunya, and yellow fever. One emerging solution to control these viruses is the release of the insect bacterium *Wolbachia*, whose infection in mosquitoes reduces virus transmission to humans. However, the effects of rising temperatures on the efficacy of this tool are unclear. Here, we studied whether DENV and *Wolbachia* can alter the thermal sensitivity of the mosquito *Ae. aegypti* by using a heat-based physiological assay. We demonstrate that, separately, DENV and *Wolbachia* infections increase mosquito thermal sensitivity, causing more rapid death when mosquitoes are exposed to extreme heat. The impacts of the microbes on mosquito thermal sensitivity were similar but not additive, suggesting they effect the mosquito in similar ways. Our work demonstrates that future global projections of DENV transmission risk and of *Wolbachia*'s potential efficacy may need to consider the impact of these microbes on vector survival.

Introduction

Mosquitoes are responsible for transmitting a diverse array of human disease-causing viruses such as Zika (ZIKV), chikungunya (CHIKV), West Nile, yellow fever, and dengue (DENV) [1, 2]. The most prevalent of these viruses is DENV. Transmitted by the mosquito *Aedes aegypti*, DENV is responsible for an estimated 390 million cases of dengue fever globally each year [3, 4]. While usually associated with a self-limiting febrile illness, DENV can also cause severe disease that may result in death [4–6]. Without effective antiviral drugs or a vaccine for DENV, ZIKV, or CHIKV, vector control has remained the primary strategy for reducing the incidence of these human diseases [7, 8]. Traditionally, such strategies have relied on insecticide use and larval habitat reduction. A more recent and promising approach involves the use of the insect endosymbiotic bacterium *Wolbachia pipientis* [9]. This self-spreading, vertically inherited bacterium has been transinfected into *Ae. aegypti*, where it is being released globally into wild populations for biological control (biocontrol) because *Wolbachia* limits the replication of viruses inside the mosquito, including DENV [10, 11].

Ae. aegypti is a highly anthropophilic species, restricted to regions with human settlements, where it breeds in human-made containers inside and near housing. The increasing incidence of dengue fever globally is in part due to the ever-expanding geographic range of the vector [5, 12]. Aided by increasing urbanization and climate change, 50% of the world's population is expected to live in association with *Ae. aegypti* by 2050 [13]. Changes in global temperature will shift the map of dengue fever in the following two ways: by increasing risk in previously temperate areas and reducing risk in some regions that exceed mosquito thermal optima [14–16]. The operative temperature range for *Ae. aegypti* is between 15.0 and 35.0°C [15], and temperatures outside this range can cause reductions in survival and reproduction and can impact developmental time between stages (i.e., eggs, larva, pupae) [17]. Because the mosquito body temperature is entirely dependent on their environment, they are highly susceptible to various aspects of thermal stress [16]. Thermal stress can be triggered in response to rising average temperatures, as well as extreme climatic events such as heat spikes [18] that are expected to result from greater climate variability.

The individual thermal optima of DENV and *Wolbachia* will also affect global distributions of disease. Warmer ambient temperatures have been associated with increased viral replication in mosquitoes and, consequently, a shorter extrinsic incubation period [19–21]. This parameter represents the time window between when a mosquito first takes an infectious blood meal and when it can transmit virus to a human via a subsequent bite, with a shorter extrinsic

incubation period leading to greater transmission rates [22]. Laboratory studies that have reared mosquitoes under different diurnal temperature ranges, which more closely match natural conditions, have also demonstrated changes in *Ae. aegypti* susceptibility to DENV [23]. At the molecular level, higher temperatures in mosquito cell culture systems appear to increase viral attachment and entry into cells due to assistance from heat shock proteins [24–26]. All of these parameters affect viral population dynamics in the vector that in turn will affect virus transmission rates at the population level [27–29]. In *Drosophila*, higher ambient temperatures have been shown to increase *Wolbachia* replication rates too but can also become lethal depending on the strain [30–35]. *Ae. aegypti* infected with the *w*Mel strain of *Wolbachia* exhibit reduced maternal transmission rates in response to heat spikes that can lead to the production of uninfected offspring [36]. More recently, several studies in *Drosophila* species have demonstrated that *Wolbachia* infection can change the host insect's thermal preference [37], of which the directionality varies by bacterial strain [38]

More broadly, infection in invertebrates has been shown to substantially increase host or vector susceptibility to thermal stress [39-42]. A recent study conducted by Hector et al., 2019 showed that *Daphnia magna* (water flea) infected with the bacterial pathogen *Pasteuria ramosa* exhibit a reduction in thermal limits up to 2°C. In *Drosophila melanogaster*, immune activation induced by bacterial challenge was shown to affect the temperature at which physiological failure occurred, reducing the overall thermal tolerance (i.e. critical thermal maximum) of the host [39]. Parallel studies have not been carried out for the major mosquito vectors of human disease-causing viruses. In the case of *Wolbachia*, a single study in *Ae. aegypti* has revealed that exposure to heat stress made the vector susceptible to starvation in the presence of the symbiont [43]. Both DENV and *Wolbachia* are pervasive throughout mosquito tissues [44, 45] and, therefore, have substantial potential to affect host physiology either directly at the cellular level or indirectly through physiological tradeoffs resulting from activation of the vector's immune response [46, 47].

Laboratory-based physiological performance assays are commonly used in invertebrates to characterize thermal tolerance [39, 48, 49]. By use of either dynamic or static regimes, these assays provide physiological parameter estimates to inform species distribution models (SDMs). SDMs are important tools used for predicting changes in climate and the response of species and habitats to environmental perturbations. One of the climatic factors commonly used in SDMs is temperature, as it is a major determining factor in the fitness of ectotherms like insects. Critical thermal maxima experiments have been shown to effectively predict an organism's geographic range [50–54]. More specifically, studies using these experimental assays have suggested that tolerance to extreme heat events near upper critical limits is more indicative of species distributions than tolerance to average daily temperatures [50]. In this study, we used a thermal knockdown assay at a temperature near mosquito upper thermal limits to examine the impact of DENV and *Wolbachia* infection, singly and in coinfection, on *Ae. aegypti* thermal tolerance. We hypothesized that the two agents would increase mosquito sensitivity to heat. Any such effect would have the potential to mediate global disease distributions as well as the geographic range over which *Wolbachia*-based biocontrol may be effective.

Results

DENV-infected mosquitoes have greater theromosenstivity

By immersing individual mosquitoes in glass vials in water heated to 42°C, which is beyond the critical maximum temperature, and measuring their time to immobilization, we obtained a measure of 'knockdown' (KD) time (Fig 1). This assay and others like it are commonly used in



Fig 1. Experimental setup to measure thermal sensitivity of DENV- and Wolbachia-infected mosquitoes.

Drosophila species to obtain estimates of thermal sensitivity but has not previously used to study mosquitoes.

To test whether DENV and or *Wolbachia* infection alter mosquito thermal sensitivity, we submerged mosquitoes (+/-*Wolbachia*, +/-DENV) in glass vials in a water bath heated to 42°C, representing the upper critical thermal limit (CT_{max}) for the mosquitoes, as determined by pilot assays. We then visually monitored the time it took for mosquitoes to become immobilized (on their backs), and this 'knockdown' (KD) time was recorded using a barcode scanner.

First, we fed DENV to 9-day-old mated female mosquitoes via a blood meal and then allowed the virus to replicate for 8 days before performing the knockdown assay. Age-matched controls were fed virus-free blood of the same donor and stock. Mosquitoes infected with DENV showed a greater sensitivity to heat (Fig 2) (according to a generalized linear model; S1 Table: 'DENV Infection': F = 22.46, df = 2, p < 0.0001), an effect observed to vary among temporal replicates ("Replicate": F = 5.07, df = 3, p = 0.035). On average, DENV-infected mosquitoes had a median KD time that was 2.9-fold faster than that of DENV-uninfected mosquitoes. Tukey's post-hoc comparisons indicated that the impact of DENV was significant (p < 0.0001) for 2 of the 4 temporal replicates (S2 Table). As genetic and environmental effects should be uniform across replicates, differences in the impact of DENV could be explained by day/circadian rhythms. There was, however, no clear trend of KD time decreasing or increasing with time of day/sequential replicate.

Viral load does not determine time to thermal knockdown

We then examined whether there was a relationship between viral load and time to knockdown (Fig 3), as pathogen load often predicts virulence in many systems, including DENV [55]. Surprisingly, we saw no such relationship between dengue load and knockdown time in our) WT+(Wildtype) line (Pearson correlation, df = 57, p>0.05) despite that mosquito body loads ranged from 10⁶ to 10⁸ viral genome copies/mosquito.

Thermal limits under viral and bacterial infection

Several *Wolbachia* strains are being released globally [56, 57]. Here, we used *w*AlbB, a strain transferred from *Aedes albopictus* [10] into *Ae. aegypti* that has shown promise with respect to



Fig 2. Thermal limits under viral infection. Knockdown time is expressed in seconds for DENV-infected (D+) and DENV-uninfected (D-) mosquitoes with no *Wolbachia* present in either treatment. Each replicate (4) contained 20 individuals per treatment (40 total per block). Box plots represent individual replicate medians and confidence intervals. Both the factors 'DENV Infection' (p<0.001) and 'Replicate' (p = 0.035) were significant by ANOVA. *p*-values report Tukey's post hoc comparison for each replicate.

reducing incidence of dengue fever following field releases in Malaysia [57]. We assessed the effect of Wolbachia infection (wAlbB) vs. wildtype on mosquito knockdown time in association with DENV infection (+/-) (Fig 4). In our generalized linear model (DENV + Wolbachia + DENV: Wolbachia + Rep), 'DENV infection' (F = 94.64, df = 1, p < .0001), 'Wolbachia infection' (F = 23.75, df = 1, p < .0001), and 'Temporal Replicate' (F = 5.05 df = 5, p = 0.0002) were all significant, with both infections increasing thermal sensitivity or decreasing knockdown (KD) time (S3 Table). There was also a significant interaction between 'DENV infection' and 'Wolbachia infection' (F = 55.68, df = 1, p < .0001). Because 'Temporal Replicate' was significant, we then followed with individual ANOVAs for each replicate (S4 Table) so that we could carry out individual Tukey's post hoc comparisons (S5 Table). In 6/6 replicates, DENV infection significantly reduced KD time. On average, across replicates, the median KD time of DENV-infected mosquitoes (D+W) was 4.5-fold more rapid (Fig 4) than that of uninfected controls (D-W-). This knockdown (KD) appears to be faster than that seen in experiment 1 (Fig 2), but comparisons across experiments are not valid given different virus preparations and use of separate mosquito cohorts. Wolbachia infection also reduced median KD time in 4/6 replicates (#'s 2, 4–6), conferring a 2.5-fold faster average KD time than WT mosquitoes in



Fig 3. Relationship between viral load and thermal knockdown in mosquitoes. Relationship between knockdown time (seconds) and DENV load (per mosquito) in wildtype (D+/W-) mosquitoes. Each point represents a single mosquito with individuals from all 4 replicate experiments. There was no significant relationship (Pearson correlation, p>0.05) between knockdown time and DENV load for the pooled set or for individual replicate experiments.

the absence of DENV (D-*W*+ vs D-*W*-). Not surprisingly, the double-negative state (D-*W*-) has a longer KD time than D+W+ for all 6 replicates. If *Wolbachia*-mediated blocking was protecting the mosquito from DENV-induced thermal sensitivity, it would be seen in the D+W- vs D+W+ comparison. None of these comparisons were significant for any of the replicates. Additionally, if the two infections were additive, one would expect D+W+ to have a faster KD time than either D+W- or D-W+. None of these comparisons are significant for any of the



Fig 4. Impact of dual microbe infection on mosquito thermal sensitivity. The effect of *Wolbachia* and DENV infection on knockdown time (seconds) across 6 temporal replicates, each containing 10 mosquitoes per the 4 treatment combinations. Box plots represent individual replicate medians and confidence intervals for wildtype infected (D+/W-) and uninfected (D-/W-) mosquitoes, along with *Wolbachia* infected (W+) and uninfected (W-) individuals. Key Tukey's *post hoc* comparisons for each replicate are described in the text and <u>S5 Table</u>. In brief, the effects of DENV and *Wolbachia* individually were significant in all 6 replicates, as was the comparison between the single infection (DENV) and the double infection (DENV and *Wolbachia*).

https://doi.org/10.1371/journal.pntd.0009548.g004



Fig 5. Relationship between bacterial load and time to knockdown in mosquitoes. Each point represents a single mosquito, with individuals from all 4 replicate experiments presented on one graph. There was no significant relationship (Pearson correlation, p < 0.05) between knockdown time and *Wolbachia* load for the pooled set or for individual replicate experiments analyzed separately (<u>S6 Table</u>).

replicates. However, in only one replicate (#5), the DENV effect is greater than the *Wolbachia* effect (D+W- vs D-W+), increasing KD time. This pattern of DENV potentially being stronger than *Wolbachia* is also in keeping with that *Wolbachia* was significant in 5/6 ANOVAs compared to 6/6 for DENV (<u>S4 Table</u>).

Microbial load does not determine time to thermal knockdown

As in Experiment 1 above, there was no evidence that KD time was determined by DENV load in Experiment 2 (Pearson correlation, df = 98, p = 0.2048) (S1 Fig). As for DENV above, we examined whether a relationship existed between total body load of *Wolbachia* and knockdown(KD) time (Fig 5). There was no significant relationship for the data when pooled across replicate (Pearson correlation, df = 117, p>0.05) or when analyzed individually (S6 Table). Interestingly, we noted that *Wolbachia* loads were lower when DENV was present (Figs 5 and S2; F = 1.924, df = 116, p<0.0001), which may relate to a virus-induced immune killing of *Wolbachia* or resource competition between *Wolbachia* and DENV, as both microbes share similar host resources [58]. We therefore also split the mosquitoes into DENV+ or -, pooled across replicates, and retested for a correlation between *Wolbachia* load and knockdown(KD) time but saw none for either D+W+ (p>0.05, df = 57) or D-W+ (p>0.05, df = 46).

Discussion

The impact of DENV infection on vector thermal sensitivity has implications for global dengue risk under a changing climate. Numerous studies have mapped the likely range of *Ae. aegypti* into the future, based on its current occupation of global thermal zones and mechanistic effects of temperature on mosquito and pathogen traits [14, 15]. At lower temperatures, the virus may fail to replicate fast enough to traverse the mosquito body and be transmitted [59], reducing transmission risk in some areas. At slightly higher temperatures, the virus may replicate faster until reaching a performance maximum of its own [23]. Our work suggests an additional factor may affect viral success in a temperature-dependent manner—the impact on mosquito survival.

Although the assay used here represents a simplified model of heat stress, future climate models point to increasing frequencies of extreme temperature events, making short exposures to high temperatures a threat to the survival of DENV and Wolbachia infected mosquitoes. Substantial evidence from *Drosophila* and other insects suggests that their ability to survive heat stress is highly predictive of an insect's current distribution and therefore also likely its future range [48, 49]. Thermal sensitivity (KD) measures, specifically, have been shown to be a relevant proxy for fitness under field relevant conditions [50]. In agreement with this, variation seen in *Drosophila* populations for measures such as thermal knockdown (KD) time is predictive of an ability to respond to artificial selection for thermal resistance/sensitivity, and as a result single trait physiological measures like these can be used effectively for developing SDMs [51, 60].

The findings for Wolbachia wAlbB indicate that the symbiont increases mosquito thermal sensitivity, although not to the same level as DENV. Both agents infect a diversity of tissues throughout the body [11] and, although not cytotoxic, evoke an immune stress response in Ae. *aegypti* [61, 62]. Most studies suggest that DENV has little effect on host fitness except in rare cases [63] Additionally, because Wolbachia lacks a complete set of metabolic pathways, it represents an energetic drain on host resources, including amino acids [64]. Three outcomes were possible for the relationship between DENV and Wolbachia coinfection and KD time, namely, protective, additive, or similar. A protective effect might have been expected given that Wolbachia-mediated blocking is known to limit DENV loads in the body [61] (also seen in this study S3 Fig), although we did not see a correlation between *Wolbachia* and DENV loads (r = -0.052, p = 0.96). This would only be the case if increasing loads of DENV led to faster KD times, but we saw no relationship. An additive effect might have suggested that the two infectious agents acted on independent aspects of mosquito physiology and both cause thermal sensitivity. We saw no difference in thermal sensitivity between the single and doubly infected mosquitoes. We also saw no relationship between Wolbachia load and KD time, like DENV, indicating that having more of either agent did not lead to greater virulence. Taken together, our data agree with a model of the symbiont and virus acting via a shared mechanism. One possible explanation is that the microbes activate similar innate immune pathways [47, 61, 65]. The effect of this activation may have direct pleiotropic effects on thermal tolerance or act through energetic tradeoffs. Interestingly, any triggering of the heat shock response by DENV infection [62] itself was not powerful enough, or long lasting enough, to mitigate the impact of either DENV or Wolbachia on KD times.

To capture the impact of viral infection upon mosquito thermal limits, we assayed individuals in the thermal KD setup by using a static tolerance assay as compared to a dynamic assay, in which the insect is gradually exposed to ramping temperatures until thermal knockdown (KD) is achieved. However, whether these two physiological assays provide comparable measures for heat tolerance has been questioned [66]. The overall outcome for both assays is dependent on the duration of heat exposure and the temperature at which thermal stress occurs [50]. In nature, ambient temperatures are rarely constant, and mosquitoes and their pathogen are subjected to temperatures that may fluctuate throughout the day [67]. Dynamic assays have been promoted for their ecological relevance due to their gradual increase in temperature versus an acute exposure to high temperature [68, 69]. A study conducted by Rezende et al., 2014 looked at this interaction between the intensity of heat stress and exposure duration through the development of thermal tolerance landscapes (TTLs), which depict the parametrization of survival time as a function of constant temperatures plus thermal exposure duration. In a past study, this group showed that TTLs are able to predict survival in thermally variable environment's when using empirical data that have incorporated either dynamic or static measures of thermal stress.

To mediate any discrepancies between dynamic and static assays, Jørgensen et al., 2019 developed a model that allowed for the direct comparison of static and dynamic measures of heat tolerance from KD data obtained from *Drosophila* [50]. They showed that a dynamic CTmax, at any given heating rate, can be effectively modeled from static measurements like ours and used to predict species distributions [50]. Although the assay we used is a simplified representation of heat stress, future climate models point to increasing frequencies of extreme temperature events, making short exposures to high temperatures relevant. Future studies should examine these effects in the context of diurnal temperature range variability reflecting natural temperatures and repeated exposure to heat shock events.

Transient exposure exceeding the mean temperature beyond thermal optima can have costly fitness effects on an individual [70, 71]. Maximum thermal tolerance in an individual is dependent on cell performance and actions of different systems, including respiratory, circulatory, and nervous [72, 73]. High heat denatures enzymes and disturbs cellular membranes that thus impact cellar processes that the insect may rely on to function [72, 73]. Our measurement of thermal limits may have some confounding impacts on desiccation or starvation stress [74]. However, studies with Drosophila show that an exposure time of 60–90 minutes was not sufficient to trigger either a starvation or desiccation response [69]. Additionally, in our system, these responses were likely minimized because mosquitoes were provided sugar water and vials were not sealed until 25 minutes before their thermal knockdown with humid air [69]. Future research examining the interaction of multiple related stressors will provide interesting insight into the range of responses that might be seen in wild populations with more complex environments under predicted climate shifts. Additionally, our study did not allow for the host to respond behaviorally to heat stress. In the field, mosquitoes may use various responses, such seeking shade or cooler areas, when thermally stressed [75]. In general, they may also close spiracles to reduce dehydration and activate a series of pathways at the cellular level, including heat shock [62, 76].

Daily and seasonal fluctuation is higher in temperate regions, whereas tropical areas experience less seasonality [77]. Species from temperate areas and high-altitude regions have broader thermal tolerance thresholds, as they can tolerate warming due to their ability to respond to variable temperatures [78]. Tropical species like mosquitoes, however, are living close to their optimal temperatures for performance [67, 79] and experience little variation in daily and seasonal fluctuation [77, 80, 81]. The impact of global climate change on mosquito-borne disease will depend strongly on species thermal history and their overall tolerance and ability to withstand change and adapt [16, 70, 82, 83]. The context of our findings should be considered within the realm of local thermal adaption, which may lead to different responses between populations, and how they respond to thermally taxing conditions. Furthermore, environmental conditions experienced across the larval stage of the mosquito can have an impact on mosquito response to thermal stress [84]. Thermal acclimation can a happen within a population and affect subsequent adult traits that can happen irrespective of any local adaption [67, 79]. In our case, mosquitoes were lab reared at constant temperatures, but if larval rearing conditions varied within a range of thermal conditions or mosquitoes were locally adapted, we may conclude that KD could differ depending on their prior thermal exposure.

With this study, we now add increased thermal sensitivity to a list of heat-associated effects for adult *Wolbachia*-infected mosquitoes that include increased susceptibility to starvation [43], reductions in maternal transmission rates of *Wolbachia*, and loss of cytoplasmic incompatibility [36]. These temperature-associated effects may reduce the competitiveness of *Wolbachia*-infected mosquitoes in extremely hot or variable environments and affect the efficacy of the biocontrol strategy. Additionally, *Wolbachia* transinfected into mosquitoes induces fitness costs that produce bistable frequency equilibria, which limit how the symbiont spreads in

populations when established from low infection frequencies [85]. Increased thermal sensitivity may contribute to these net fitness costs and bistable equilibria. Naturally occurring *Wolbachia* infections commonly evoke a lower immune response, potentially from a longer history of coevolution with their hosts [86]. One solution for protecting *Wolbachia's* use for biocontrol in hotter regions might involve adapting mosquito: *Wolbachia* pairings to higher temperatures or for reduced immune responses in the laboratory by using artificial selection before release into the field.

Several factors may have affected our study design or should be considered in future studies. First, the effect of infection on KD time may rely heavily on the combination of *Wolbachia* strain, virus and mosquito genotypes, and their past histories of thermal adaptation. Additionally, some genotypes of both the host and pathogen may be better able to compensate for thermal stress [40]. Thermal tolerance should therefore be examined for combinations of key circulating viruses and *Wolbachia* release strains in diverse *Ae. aegypti* populations from different global regions. Second, factors like mosquito age at exposure, reproductive status, body size or nutritional status, blood meal history, and other coinfecting microbes/the microbiome may also play a role in thermal tolerance. Third, as for the Drosophila literature [48], it will be important to assess how these laboratory measures of thermal sensitivity relate to fitness measures in the field and how the interaction between viral infection and temperature affect the distribution of virus transmission. Fourth, it would be interesting to assess whether thermal sensitivity effects due to *Wolbachia* are also present in the larval stage. If so, they may affect the successful field release of *Wolbachia* via egg stage [87].

In conclusion, our work suggests that DENV- and *Wolbachia*-induced increases in mosquito thermal sensitivity may limit the geographic range of the virus's transmission to humans and the ability of the symbiont to be used for biocontrol. We suggest that future models predicting dengue distribution may also need to incorporate the interaction between virus and vector survival to be accurate, particularly at the edge of a mosquito's distribution where the potential impact of these microbes would likely be greatest in hotter and more thermally variable regions of the mosquitos' range.

Materials & methods

Mosquitoes

Within a year of this study, the *w*AlbB *Wolbachia*-infected *Ae. aegypti* line was backcrossed for 7 generations to a wild-caught mosquito line (AFM-Wildtype [WT]), that was collected from the field in Mérida, Mexico, by Pablo Manrique. This process homogenized the nuclear genetic background with the field line but would have retained the mitochondria from the *Wolbachia*-infected line. We used this continuously maintained line (~1 year) for the DENV+/- experiments. Furthermore, because this population was not naturally infected with *Wolbachia*, we also used it as a negative control for subsequent *Wolbachia* experiments. Both lines were reared under standard conditions: 12hr light/dark, 26°C, 60% relative humidity, *ad libitum* Tetramin fish food at the larval phase, and 10% sucrose as adults.

Virus

We used DENV serotype 2 for all experiments, as it has been previously shown to form strong infections in the mosquito in the laboratory [11]. Originally isolated from a patient in East Timor, the ET-300 strain (GenBank EF440433.1), with approximately 20 passages, was used in the assay, as done previously in a study of *Wolbachia*:mosquito:DENV interactions [88]. *Ae. albopictus* C6/36 cells were grown at 26°C in RPMI 1640 medium (Invitrogen, Carlsbad, CA) supplemented with 10% fetal bovine serum (FBS), 1× Glutamax (Invitrogen), and HEPES

buffer. Cells were first allowed to form monolayers of around 60–80% confluence in T-175 flasks (Sigma Aldrich, St. Louis, MO) and then were inoculated with DENV and maintained in RPMI medium supplemented with 2% FBS. At day 7 post-inoculation, live virus was harvested, titrated via absolute quantification RT-qPCR, and adjusted to a final viral load of 10⁷ DENV copies per ml.

Mosquito infections with DENV

Before an infectious blood meal, mated, 9-day-old female mosquitoes were sorted into groups of 100 in 68-oz paper cartons. Sucrose was removed from the mosquitoes 24 hrs before oral infection and replaced with water. Double-chamber glass feeders were covered with pig intestine previously immersed in a 10% sucrose solution. Water heated to 37°C was circulated in the outer chamber of the feeders, and a 1:1 mix of defibrinated human blood and the previously titrated DENV virus was placed inside the feeder. The final feed concentration was 2.5e⁷ DENV copies/ml. In parallel, all DENV– mosquitoes were fed a solution containing a 1:1 ratio of blood without virus and RPMI 1640 cell culture media to serve as mock controls. After 24 hrs, all blood fed mosquitoes were identified by visual inspection under chilling and returned to the cartons.

Thermal knockdown assay

The thermal sensitivity of infected and uninfected mosquitoes was measured using a static heat shock assay (Fig 1) based on previous work in *Drosophila* [89]. KD experiments were carried out at 42°C, as pilot studies indicated this temperature represented the critical thermal maximum of the mosquitoes (S7 Table). Exposure to 42°C led to death in 97% of individuals trialed in the WT line even without the presence of DENV. From the literature, this temperature also represents some of the upper thermal extremes insects may encounter in nature due to global climate change [17]. Mosquitoes were moved 24 hrs after blood feeding to individual 40-ml glass vials with mesh lids topped with cotton balls soaked in 10% sucrose that were changed daily. All vials were housed in environmental chambers maintained at 26°C and 65% relative humidity. Knockdown (KD) assays were performed 8 days after blood feed, allowing time for the virus to disseminate throughout the body and affect mosquito physiology [19]. Before each assay (<25 minutes), the mesh lids to the vials were replaced with solid plastic. The assays themselves were also carried out in the environmental chambers, so air captured in the vials upon sealing was at 65% relative humidity. The vials containing the mosquitoes were then attached to a plastic board in groups of 40 via anchored clips, randomized with respect to treatment. The board with vials was immersed in a water bath heated to 42°C and allowed a 60-second acclimation period. Mosquitoes were then monitored visually for immobility and time to thermal knockdown (KD) was scored using Brady labels and a TriCoder Scanner (Worth Data Inc., Santa Cruz, California). Immobility was confirmed by tapping on the vial while it was still immersed in the water bath. Mosquitoes did not recover after thermal knockdown(KD). The DENV status of all individuals was confirmed by PCR as described below.

Mosquito nucleic acid extraction

Upon completion of each KD experiment, individual whole mosquitoes were anesthetized by chilling and placed in 1.5-ml microfuge tubes (Sarstedt, Nümbrecht, Germany) containing 300 µl of TRIzol reagent (Invitrogen, Carlsbad, CA, USA) and a 2.8-mm ceramic bead. Samples were homogenized on a Bead Ruptor Elite (Omni International, USA) and then frozen at -80°C. Total RNA was extracted with the Direct-zol RNA 96 Magbead Zymo kit (Zymo Research, Irvine, CA) according to the manufacturer's protocol. Following this step, the

samples were processed using an automatic magnetic bead purification system (MagMAX Express 96 system, Applied Biosystems). RNA was eluted in 50 µl RNase free water. RNA was then treated with 5 units of DNase I (Sigma-Aldrich) at room temperature for 15 min, followed by inactivation with 50 mM EDTA at 70°C for 10 min. To measure *Wolbachia* loads, extractions for both RNA and DNA were performed using the column-based Direct-zol DNA/ RNA Miniprep kit. RNA was eluted in 50 µl RNase free water, followed by DNA elution in 50 µl of Direct-zol DNA Elution Buffer. Total RNA and DNA concentrations were determined with a NanoDrop model 2000/2000C (Thermo Scientific, Waltham, MA).

DENV quantification

DENV virus was quantified using TaqMan Fast Virus 1-step Master Mix (Thermo Fisher Scientific) in 10-µl reaction volumes with DENV-specific primers and probes [90] (S8 Table). The following protocol was used: reverse transcription at 50°C for 5 min, followed by 50 amplification cycles 95°C for 20 sec, and amplification cycling at 95°C for 3 sec and 60°C for 30 sec. A standard reference curve of known quantities of a DENV-2 genomic fragment was used for absolute quantification by qPCR. The DENV-2 genomic fragment was previously inserted into a plasmid and transformed into *Escherichia coli* as described [90]. The linearized and purified fragment was serially diluted ranging from 10^7 to 10^2 copies and used to create a standard curve of DENV amplification. The standard curve was run in duplicate on each 96-well plate, and the limit of detection was set at 10^2 copies. All samples were run in duplicate.

Wolbachia quantification

Wolbachia load was assessed as previously reported including published primers and probes [90]. In brief, a multiplex qPCR reaction amplifying the target *Wolbachia*-specific surface protein *wsp* and mosquito ribosome subunit 17 housekeeping gene *RpS17* was performed. The *RpS17* gene was used to normalize *wsp* gene copies. Quantitative PCR reactions were run in duplicate and performed in a 10-µl total volume containing 1× Lightcycler 480 Probes Master reaction mix, 5 µM of each *wsp* primers and probe, 2.5 µM each of *RpS17* primers and probe, and 1 µl of DNA template. Cycling was performed using a LightCycler480 Instrument (Roche), with 1 cycle at 95°C for 5 min; followed by 45 amplification cycles of 95°C for 10 s, 60°C for 15 s, and 72°C for 1 s; and a final cooling cycle of 40°C for 10 s. Target to housekeeping gene ratios were calculated using the Livak's $2^{-\Delta\Delta CT}$ method relative quantification algorithm in the Lightcycler 480 software (Roche).

Experimental design and statistical analysis

In the first set of experiments, mosquitoes without *Wolbachia* were infected with DENV (passage 36) as described above and tested for KD in 4 temporal replicates each containing 20 individual mosquitoes per treatment (+/–DENV). All mosquitoes were from the same generation/ population of mosquitoes. In the second set of experiments, we explored both *Wolbachia* (+/-) and DENV (+/-) infection. *Wolbachia*-free mosquitoes were drawn from the same population as above, 3 generations hence. As for the first experiment, live virus was cultured to feed to these mosquitoes, from virus passaged 38 times. Any mosquitoes not infected with DENV in DENV+ treatments (~4% in *W*- mosquitoes in experiments 1 and 2 and 53% *W*+ mosquitoes in experiment 2) were excluded from the data analysis. All knockdown data were log transformed before analysis to correct issues with skew. In experiment 2, each of the 4 combinations of infection status was represented by 10 individual mosquitoes in a temporal replicate. These experiments were then replicated 6 times (blocks). All statistics were carried out using JMP Version 4 for Mac (SAS Institute Inc., Cary, NC, USA). Knockdown (KD) time was examined using general linear models with 'DENV infection,' 'temporal replicate,' and '*Wolbachia* infection' as fixed effects where relevant. The relationships between each DENV load and *Wolbachia* load and KD time were examined with Pearson's correlation.

Supporting information

S1 Table. Impact of DENV infection alone on KD time. ANOVA for <u>Fig 2</u>, with 'DENV infection status' and 'temporal replicate' as factors. (DOCX)

S2 Table. Impact of DENV infection alone on KD time for each replicate independently. Tukey's post hoc comparisons for <u>Fig 2</u> for DENV infection status (D+/-). (DOCX)

S3 Table. Impact of DENV and Wolbachia co-infection on KD time. GLM for KD time for Fig 4 including 'Rep', 'DENV status', and '*Wolbachia* status' as factors. (DOCX)

S4 Table. Impact of DENV and *Wolbachia* co-infection on KD time for each replicate independently. ANOVAs for knockdown time for Fig <u>4</u> for each of the six individual temporal replicates with 'DENV status' and 'Wolbachia status' as factors. (DOCX)

S5 Table. Impact of DENV and *Wolbachia* infections in mono and co-infection on KD time for each replicate independently. Tukey's post hoc comparisons of KD time for <u>Fig 4</u> for all combinations of DENV (D+/-) and *Wolbachia* (W+/-) infection status. (DOCX)

S6 Table. Correlation between *Wolbachia* **load and KD time.** Summary of individual rep correlations between *Wolbachia* load and KD time for <u>Fig 4</u>. (DOCX)

S7 Table. Survival during pilot knockdowns at varying temperatures (DENV+/-). Summary of t-tests from pilot experiments conducted at varying temperatures showing average time to knockdown for dengue infected (D+*W*-) and uninfected (D-*W*-) individuals. All temperatures resulted in a knockdown phenotype, but 42°C led to death in 97% of individuals trialed in the WT line even without the presence of DENV. 42°C also allowed resulted in faster knockdown times during the assay.

(DOCX)

S8 Table. Primers and probes for dengue virus serotype 2 detection. (DOCX)

S1 Fig. Relationship between viral load and thermal knockdown in mosquitoes for <u>Fig 4</u>. KD time (seconds) vs. DENV load (per mosquito) in Wildtype (*W*-) mosquitoes. Each point represents a single mosquito. Data include all individuals across 6 replicate experiments. There was no significant relationship (Pearson's correlation, P = 0.204) between KD time and DENV load.

(DOCX)

S2 Fig. Reduction of *Wolbachia* **load in the presence of DENV for** <u>Fig 4</u>. Average *Wolbachia* loads with (D+W+) and without (D-W+) DENV. Data are pooled across 6 replicate experiments. Graphs depict mean \pm sem *Wolbachia* per host cell. *Wolbachia* load is reduced in the

presence of DENV infection (df = 116, F-Ratio = 1.92, p<0.0001). (DOCX)

S3 Fig. Evidence of *Wolbachia*-mediated blocking of DENV for Fig 4. Average DENV loads for *Wolbachia* infected (D+W+) and uninfected (D+W-). Data are pooled across the 6 replicate experiments. Graphs depict mean \pm standard error. Loads are reduced in the presence of *Wolbachia* (df = 115, F = 13.32, p<0.001). (DOCX)

Acknowledgments

The authors would like to thank Michael D. Cannon and Austin Mejia for their assistance with knockdown (KD) assays and Pablo Manrique-Saide for the wildtype *Ae. aegypti* line.

Author Contributions

- **Conceptualization:** Fhallon Ware-Gilmore, Carla M. Sgrò, Zhiyong Xi, Katriona Shea, Matthew D. Hall, Matthew B. Thomas, Elizabeth A. McGraw.
- **Formal analysis:** Fhallon Ware-Gilmore, Carla M. Sgrò, Heverton L. C. Dutra, Katriona Shea, Matthew D. Hall, Elizabeth A. McGraw.
- **Funding acquisition:** Carla M. Sgrò, Matthew D. Hall, Matthew B. Thomas, Elizabeth A. McGraw.

Investigation: Fhallon Ware-Gilmore, Matthew J. Jones.

Methodology: Fhallon Ware-Gilmore, Heverton L. C. Dutra, Matthew J. Jones.

Project administration: Elizabeth A. McGraw.

Resources: Zhiyong Xi.

Supervision: Matthew B. Thomas, Elizabeth A. McGraw.

Writing - original draft: Fhallon Ware-Gilmore, Matthew B. Thomas, Elizabeth A. McGraw.

Writing – review & editing: Fhallon Ware-Gilmore, Carla M. Sgrò, Zhiyong Xi, Heverton L. C. Dutra, Matthew J. Jones, Katriona Shea, Matthew D. Hall, Elizabeth A. McGraw.

References

- Halstead SB. Dengue virus–mosquito Interactions. Annu Rev Entomol. 2008; 53: 273–291. <u>https://doi.org/10.1146/annurev.ento.53.103106.093326</u> PMID: <u>17803458</u>
- Qureshi AI, editor. Chapter 2—Mosquito-borne diseases. Zika virus disease. Academic Press; 2018. pp. 27–45. <u>https://doi.org/10.1016/B978-0-12-812365-2.00003–2</u>
- Souza-Neto JA, Powell JR, Bonizzoni M. Aedes aegypti vector competence studies: A review. Infect Genet Evol. 2019; 67: 191–209. <u>https://doi.org/10.1016/j.meegid.2018.11.009</u> PMID: <u>30465912</u>
- Brady OJ, Gething PW, Bhatt S, Messina JP, Brownstein JS, Hoen AG, et al. Refining the global spatial limits of dengue virus transmission by evidence-based consensus. Reithinger R, editor. PLOS Negl Trop Dis. 2012; 6: e1760. <u>https://doi.org/10.1371/journal.pntd.0001760</u> PMID: <u>22880140</u>
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, Moyes CL, et al. The global distribution and burden of dengue. Nature. 2013; 496: 504–507. <u>https://doi.org/10.1038/nature12060</u> PMID: 23563266
- Messina JP, Brady OJ, Pigott DM, Brownstein JS, Hoen AG, Hay SI. A global compendium of human dengue virus occurrence. Sci Data. 2014; 1: 1–6. <u>https://doi.org/10.1038/sdata.2014.4</u> PMID: 25977762
- McArthur MA, Sztein MB, Edelman R. Dengue vaccines: recent developments, ongoing challenges and current candidates. Expert Rev Vaccines. 2013; 12: 933–953. <u>https://doi.org/10.1586/14760584.2013.</u> 815412 PMID: 23984962

- Rather IA, Parray HA, Lone JB, Paek WK, Lim J, Bajpai VK, et al. Prevention and control strategies to counter dengue virus infection. Front Cell Infect Microbiol. 2017; 7. <u>https://doi.org/10.3389/fcimb.2017</u>. 00336 PMID: 28791258
- Flores HA, O'Neill SL. Controlling vector-borne diseases by releasing modified mosquitoes. Nat Rev Microbiol. 2018; 16: 508–518. <u>https://doi.org/10.1038/s41579-018-0025-0</u> PMID: 29777177
- Xi Z, Khoo CCH, Dobson SL. Wolbachia establishment and invasion in an Aedes aegypti laboratory population. Science. 2005; 310: 326–328. https://doi.org/10.1126/science.1117607 PMID: 16224027
- Walker T, Johnson PH, Moreira LA, Iturbe-Ormaetxe I, Frentiu FD, McMeniman CJ, et al. The *w*Mel Wolbachia strain blocks dengue and invades caged Aedes aegypti populations. Nature. 2011; 476: 450–453. https://doi.org/10.1038/nature10355 PMID: 21866159
- Brady OJ, Hay SI. The Global Expansion of Dengue: How Aedes aegypti mosquitoes enabled the first pandemic arbovirus. Annu Rev Entomol. 2020; 65: 191–208. <u>https://doi.org/10.1146/annurev-ento-011019-024918 PMID: 31594415</u>
- Kraemer MUG, Reiner RC, Brady OJ, Messina JP, Gilbert M, Pigott DM, et al. Past and future spread of the arbovirus vectors *Aedes aegypti* and *Aedes albopictus*. Nat Microbiol. 2019; 4: 854–863. <u>https://doi.org/10.1038/s41564-019-0376-y</u> PMID: <u>30833735</u>
- Ryan SJ, Carlson CJ, Mordecai EA, Johnson LR. Global expansion and redistribution of Aedes-borne virus transmission risk with climate change. Han BA, editor. PLOS Negl Trop Dis. 2019; 13: e0007213. <u>https://doi.org/10.1371/journal.pntd.0007213</u> PMID: <u>30921321</u>
- Mordecai EA, Cohen JM, Evans MV, Gudapati P, Johnson LR, Lippi CA, et al. Detecting the impact of temperature on transmission of Zika, dengue, and chikungunya using mechanistic models. PLOS Negl Trop Dis. 2017; 11: e0005568. https://doi.org/10.1371/journal.pntd.0005568 PMID: 28448507
- Mordecai EA, Caldwell JM, Grossman MK, Lippi CA, Johnson LR, Neira M, et al. Thermal biology of mosquito-borne disease. Ecol Lett. 2019; 22: 1690–1708. <u>https://doi.org/10.1111/ele.13335</u> PMID: <u>31286630</u>
- Reinhold J, Lazzari C, Lahondère C. Effects of the Environmental Temperature on Aedes aegypti and Aedes albopictus mosquitoes: a review. Insects. 2018; 9: 158. <u>https://doi.org/10.3390/insects9040158</u> PMID: <u>30404142</u>
- Jia P, Liang L, Tan X, Chen J, Chen X. Potential effects of heat waves on the population dynamics of the dengue mosquito *Aedes albopictus*. Yakob L, editor. PLOS Negl Trop Dis. 2019; 13: e0007528. https://doi.org/10.1371/journal.pntd.0007528 PMID: 31276467
- 19. Chan M, Johansson MA. The incubation periods of dengue viruses. PLOS One. 2012; 7: e50972. https://doi.org/10.1371/journal.pone.0050972 PMID: 23226436
- Rohani A, Wong YC, Zamre I, Lee HL, Zurainee MN. The effect of extrinsic incubation temperature on development of dengue serotype 2 and 4 viruses in *Aedes aegypti* (L.). Southeast Asian J Trop Med Public Health. 2009; 40: 942–950. PMID: <u>19842378</u>
- Kamiya T, Greischar MA, Wadhawan K, Gilbert B, Paaijmans K, Mideo N. Temperature-dependent variation in the extrinsic incubation period elevates the risk of vector-borne disease emergence. Epidemics. 2020; 30: 100382. <u>https://doi.org/10.1016/j.epidem.2019.100382</u> PMID: <u>32004794</u>
- Liu Z, Zhang Z, Lai Z, Zhou T, Jia Z, Gu J, et al. Temperature increase enhances Aedes albopictus competence to transmit dengue virus. Front Microbiol. 2017; 8. <u>https://doi.org/10.3389/fmicb.2017.02337</u> PMID: 29250045
- Lambrechts L, Paaijmans KP, Fansiri T, Carrington LB, Kramer LD, Thomas MB, et al. Impact of daily temperature fluctuations on dengue virus transmission by *Aedes aegypti*. Proc Natl Acad Sci. 2011; 108: 7460–7465. https://doi.org/10.1073/pnas.1101377108 PMID: 21502510
- Pujhari S, Brustolin M, Macias VM, Nissly RH, Nomura M, Kuchipudi SV, et al. Heat shock protein 70 (Hsp70) mediates Zika virus entry, replication, and egress from host cells. Emerg Microbes Infect. 2019; 8: 8–16. <u>https://doi.org/10.1080/22221751.2018.1557988</u> PMID: <u>30866755</u>
- Kanlaya R, Thongboonkerd V. Heat shock protein 70 (HSP70) family in dengue virus infection. In: Asea AAA, Kaur P, editors. Regulation of heat shock protein responses. Cham: Springer International Publishing; 2018. pp. 395–409. <u>https://doi.org/10.1007/978-3-319-74715-6_16</u>
- Vancini R, Wang G, Ferreira D, Hernandez R, Brown DT. Alphavirus genome delivery occurs directly at the plasma membrane in a time- and temperature-dependent process. J Virol. 2013; 87: 4352–4359. <u>https://doi.org/10.1128/JVI.03412-12</u> PMID: 23388718
- Wilke ABB, Medeiros-Sousa AR, Ceretti-Junior W, Marrelli MT. Mosquito populations dynamics associated with climate variations. Acta Trop. 2017; 166: 343–350. <u>https://doi.org/10.1016/j.actatropica.2016</u>. 10.025 PMID: 27810426

- Marini G, Poletti P, Giacobini M, Pugliese A, Merler S, Rosà R. The role of climatic and density dependent factors in shaping mosquito population dynamics: the case of *Culex pipiens* in Northwestern Italy. PLOS ONE. 2016; 11: e0154018. https://doi.org/10.1371/journal.pone.0154018 PMID: 27105065
- Scott TW, Amerasinghe PH, Morrison AC, Lorenz LH, Clark GG, Strickman D, et al. Longitudinal studies of Aedes aegypti (Diptera: Culicidae) in Thailand and Puerto Rico: blood feeding frequency. J Med Entomol. 2000; 37: 89–101. https://doi.org/10.1603/0022-2585-37.1.89 PMID: 15218911
- Clancy DJ, Hoffmann AA. Environmental effects on cytoplasmic incompatibility and bacterial load in Wolbachia-infected Drosophila simulans. Entomol Exp Appl. 1998; 86: 13–24. <u>https://doi.org/10.1046/j. 1570-7458.1998.00261.x</u>
- Hurst GD, Johnson AP, Schulenburg JH, Fuyama Y. Male-killing Wolbachia in Drosophila: a temperature-sensitive trait with a threshold bacterial density. Genetics. 2000; 156: 699–709. PMID: <u>11014817</u>
- Mouton L, Henri H, Bouletreau M, Vavre F. Effect of temperature on *Wolbachia* density and impact on cytoplasmic incompatibility. Parasitol. 2006; 132: 49–56. <u>https://doi.org/10.1017/S0031182005008723</u> PMID: 16393353
- **33.** Strunov A, Kiseleva E, Gottlieb Y. Spatial and temporal distribution of pathogenic *Wolbachia* strain *w*MelPop in *Drosophila melanogaster* central nervous system under different temperature conditions. J Invertebr Pathol. 2013; 114: 22–30. https://doi.org/10.1016/j.jip.2013.05.001 PMID: 23672808
- Correa CC, Ballard JWO. Wolbachia gonadal density in female and male Drosophila vary with laboratory adaptation and respond differently to physiological and environmental challenges. J Invertebr Pathol. 2012; 111: 197–204. <u>https://doi.org/10.1016/j.jip.2012.08.003</u> PMID: <u>22903036</u>
- Wiwatanaratanabutr I, Kittayapong P. Effects of crowding and temperature on Wolbachia infection density among life cycle stages of Aedes albopictus. J Invertebr Pathol. 2009; 102: 220–224. <u>https://doi.org/10.1016/j.jip.2009.08.009</u> PMID: <u>19686755</u>
- Ross PA, Axford JK, Yang Q, Staunton KM, Ritchie SA, Richardson KM, et al. Heatwaves cause fluctuations in *w*Mel *Wolbachia* densities and frequencies in *Aedes aegypti*. PLOS Negl Trop Dis. 2020; 14: e0007958. https://doi.org/10.1371/journal.pntd.0007958 PMID: 31971938
- Truitt AM, Kapun M, Kaur R, Miller WJ. Wolbachia modifies thermal preference in *Drosophila melanoga-ster*. Environ Microbiol. 2019; 21: 3259–3268. <u>https://doi.org/10.1111/1462-2920.14347</u> PMID: 29971900
- Hague MTJ, Caldwell CN, Cooper BS. Pervasive effects of Wolbachia on host temperature preference. mBio. 2020; 11. <u>https://doi.org/10.1128/mBio.01768-20</u> PMID: <u>33024036</u>
- Hector TE, Sgrò CM, Hall MD. The influence of immune activation on thermal tolerance along a latitudinal cline. J Evol Biol. 2020; jeb.13663. <u>https://doi.org/10.1111/jeb.13663</u> PMID: <u>32506574</u>
- Hector TE, Sgrò CM, Hall MD. Pathogen exposure disrupts an organism's ability to cope with thermal stress. Glob Change Biol. 2019; 25: 3893–3905. https://doi.org/10.1111/gcb.14713 PMID: 31148326
- Ross PA, Ritchie SA, Axford JK, Hoffmann AA. Loss of cytoplasmic incompatibility in *Wolbachia*infected *Aedes aegypti* under field conditions. PLOS Negl Trop Dis. 2019; 13: e0007357. <u>https://doi.org/10.1371/journal.pntd.0007357 PMID: 31002720</u>
- 42. Porras MF, Navas CA, Marden JH, Mescher MC, De Moraes CM, Pincebourde S, et al. Enhanced heat tolerance of viral-infected aphids leads to niche expansion and reduced interspecific competition. Nat Commun. 2020; 11: 1184. https://doi.org/10.1038/s41467-020-14953-2 PMID: 32132537
- 43. Foo IJ-H, Hoffmann AA, Ross PA. Cross-generational effects of heat stress on fitness and Wolbachia density in Aedes aegypti mosquitoes. Trop Med Infect Dis. 2019;4. <u>https://doi.org/10.3390/tropicalmed4010013</u> PMID: <u>30642130</u>
- Amuzu HE, McGraw EA. Wolbachia-based dengue virus inhibition is not tissue-specific in Aedes aegypti. PLOS Negl Trop Dis. 2016; 10. https://doi.org/10.1371/journal.pntd.0005145 PMID: 27855218
- Novelo M, Hall MD, Pak D, Young PR, Holmes EC, McGraw EA. Intra-host growth kinetics of dengue virus in the mosquito *Aedes aegypti*. PLOS Pathog. 2019; 15: e1008218. <u>https://doi.org/10.1371/journal.ppat.1008218</u> PMID: <u>31790509</u>
- 46. Mukherjee D, Das S, Begum F, Mal S, Ray U. The mosquito immune system and the life of dengue virus: what we know and do not know. Pathogens. 2019; 8. <u>https://doi.org/10.3390/pathogens8020077</u> PMID: <u>31200426</u>
- Ramirez JL, Dimopoulos G. The Toll immune signaling pathway control conserved anti-dengue defenses across diverse *Ae. aegypti* strains and against multiple dengue virus serotypes. Dev Comp Immunol. 2010; 34: 625–629. https://doi.org/10.1016/j.dci.2010.01.006 PMID: 20079370
- Overgaard J, Kearney MR, Hoffmann AA. Sensitivity to thermal extremes in Australian *Drosophila* implies similar impacts of climate change on the distribution of widespread and tropical species. Glob Change Biol. 2014; 20: 1738–1750. <u>https://doi.org/10.1111/gcb.12521</u> PMID: <u>24549716</u>

- 49. Sgrò CM, Overgaard J, Kristensen TN, Mitchell KA, Cockerell FE, Hoffmann AA. A comprehensive assessment of geographic variation in heat tolerance and hardening capacity in populations of Drosophila melanogaster from eastern Australia. J Evol Biol. 2010; 23: 2484–2493. <u>https://doi.org/10.1111/j. 1420-9101.2010.02110.x</u> PMID: <u>20874849</u>
- Jørgensen LB, Malte H, Overgaard J. How to assess Drosophila heat tolerance: Unifying static and dynamic tolerance assays to predict heat distribution limits. Funct Ecol. 2019; 33: 629–642. <u>https://doi.org/10.1111/1365-2435.13279</u>
- Kearney M, Porter W. Mechanistic niche modelling: combining physiological and spatial data to predict species' ranges. Ecol Lett. 2009; 12: 334–350. <u>https://doi.org/10.1111/j.1461-0248.2008.01277.x</u> PMID: <u>19292794</u>
- Rezende EL, Bozinovic F, Szilágyi A, Santos M. Predicting temperature mortality and selection in natural Drosophila populations. Science. 2020; 369: 1242–1245. <u>https://doi.org/10.1126/science.aba9287</u> PMID: 32883867
- Hoffmann AA. Physiological climatic limits in Drosophila: patterns and implications. J Exp Biol. 2010; 213: 870–880. https://doi.org/10.1242/jeb.037630 PMID: 20190112
- Kellermann V, Overgaard J, Hoffmann AA, Fløjgaard C, Svenning J-C, Loeschcke V. Upper thermal limits of Drosophila are linked to species distributions and strongly constrained phylogenetically. Proc Natl Acad Sci. 2012; 109: 16228–16233. https://doi.org/10.1073/pnas.1207553109 PMID: 22988106
- Rico-Hesse R. Dengue virus virulence and transmission determinants. Dengue Virus. 2009; 338: 45– 55. https://doi.org/10.1007/978-3-642-02215-9_4
- 56. Ryan PA, Turley AP, Wilson G, Hurst TP, Retzki K, Brown-Kenyon J, et al. Establishment of *w*Mel *Wolbachia* in *Aedes aegypti* mosquitoes and reduction of local dengue transmission in Cairns and surrounding locations in northern Queensland, Australia. Gates Open Res. 2019; 3: 1547. <u>https://doi.org/10.12688/gatesopenres.13061.2 PMID: 31667465</u>
- Nazni WA, Hoffmann AA, NoorAfizah A, Cheong YL, Mancini MV, Golding N, et al. Establishment of Wolbachia strain wAlbB in Malaysian populations of Aedes aegypti for dengue control. Curr Biol. 2019; 29: 4241–4248.e5. https://doi.org/10.1016/j.cub.2019.11.007 PMID: 31761702
- Koh C, Islam MN, Ye YH, Chotiwan N, Graham B, Belisle JT, et al. Dengue virus dominates lipid metabolism modulations in *Wolbachia*-coinfected *Aedes aegypti*. Commun Biol. 2020; 3: 1–14. <u>https://doi.org/10.1038/s42003-019-0734-6</u> PMID: <u>31925316</u>
- Alto BW, Bettinardi D. Temperature and dengue virus infection in osquitoes: independent effects on the immature and adult stages. Am J Trop Med Hyg. 2013; 88: 497–505. <u>https://doi.org/10.4269/ajtmh.12-0421</u> PMID: 23382163
- Blackburn S, Heerwaarden B van, Kellermann V, Sgrò CM. Evolutionary capacity of upper thermal limits: beyond single trait assessments. J Exp Biol. 2014; 217: 1918–1924. <u>https://doi.org/10.1242/jeb.</u> 099184 PMID: 24625644
- Pan X, Zhou G, Wu J, Bian G, Lu P, Raikhel AS, et al. *Wolbachia* induces reactive oxygen species (ROS)-dependent activation of the Toll pathway to control dengue virus in the mosquito *Aedes aegypti*. Proc Natl Acad Sci U S A. 2012; 109: E23–31. <u>https://doi.org/10.1073/pnas.1116932108</u> PMID: 22123956
- Sigle LT, McGraw EA. Expanding the canon: Non-classical mosquito genes at the interface of arboviral infection. Insect Biochem Mol Biol. 2019; 109: 72–80. <u>https://doi.org/10.1016/j.ibmb.2019.04.004</u> PMID: 30970277
- Sylvestre G, Gandini M, Maciel-de-Freitas R. Age-dependent effects of oral infection with dengue virus on *Aedes aegypti* (Diptera: Culicidae) feeding behavior, survival, oviposition success and fecundity. Coffey LL, editor. PLoS ONE. 2013; 8: e59933. <u>https://doi.org/10.1371/journal.pone.0059933</u> PMID: 23555838
- Wu M, Sun LV, Vamathevan J, Riegler M, Deboy R, Brownlie JC, et al. Phylogenomics of the reproductive parasite *Wolbachia pipientis w*Mel: a streamlined genome overrun by obile genetic elements. PLOS Biol. 2004; 2: e69. https://doi.org/10.1371/journal.pbio.0020069 PMID: 15024419
- Klepsatel P, Gáliková M, Xu Y, Kühnlein RP. Thermal stress depletes energy reserves in Drosophila. Sci Rep. 2016; 6: 33667. https://doi.org/10.1038/srep33667 PMID: 27641694
- 66. Winther Bak C, Bahrndorff S, Krog Noer N, Bjerregaard Jørgensen L, Overgaard J, Nygaard Kristensen T. Comparison of static and dynamic assays when quantifying thermal plasticity of Drosophilids. Insects. 2020; 11. <u>https://doi.org/10.3390/insects11080537</u> PMID: <u>32824251</u>
- Carrington LB, Armijos MV, Lambrechts L, Barker CM, Scott TW. Effects of fluctuating daily temperatures at critical thermal extremes on *Aedes aegypti* life-history traits. PLOS ONE. 2013; 8: e58824. <u>https://doi.org/10.1371/journal.pone.0058824</u> PMID: 23520534

- Chown SL, Jumbam KR, Sørensen JG, Terblanche JS. Phenotypic variance, plasticity and heritability estimates of critical thermal limits depend on methodological context. Funct Ecol. 2009; 23: 133–140. <u>https://doi.org/10.1111/j.1365-2435.2008.01481.x</u>
- 69. Overgaard J, Kristensen TN, Sørensen JG. Validity of thermal ramping assays used to assess thermal tolerance in arthropods. PLOS ONE. 2012; 7: e32758. <u>https://doi.org/10.1371/journal.pone.0032758</u> PMID: 22427876
- 70. Paaijmans KP, Heinig RL, Seliga RA, Blanford JI, Blanford S, Murdock CC, et al. Temperature variation makes ectotherms more sensitive to climate change. Glob Change Biol. 2013; 19: 2373–2380. <u>https:// doi.org/10.1111/gcb.12240</u> PMID: 23630036
- Thomas MB, Blanford S. Thermal biology in insect-parasite interactions. Trends Ecol Evol. 2003; 18: 344–350. <u>https://doi.org/10.1016/S0169-5347(03)00069-7</u>
- 72. Neven LG. Physiological responses of insects to heat. Postharvest Biol Technol. 2000; 21: 103–111. https://doi.org/10.1016/S0925-5214(00)00169-1
- Colinet H, Sinclair BJ, Vernon P, Renault D. Insects in fluctuating thermal environments. Annu Rev Entomol. 2015; 60: 123–140. <u>https://doi.org/10.1146/annurev-ento-010814-021017</u> PMID: 25341105
- 74. Hadley NF. Water relations of terrestrial arthropods. CUP Archive; 1994.
- 75. Kirby MJ, Lindsay SW. Responses of adult mosquitoes of two sibling species, *Anopheles arabiensis* and *A. gambiae* s.s. (Diptera: Culicidae), to high temperatures. Bull Entomol Res. 2004; 94: 441–448. https://doi.org/10.1079/ber2004316 PMID: 15385063
- 76. Arcaz AC, Huestis DL, Dao A, Yaro AS, Diallo M, Andersen J, et al. Desiccation tolerance in Anopheles coluzzii: the effects of spiracle size and cuticular hydrocarbons. J Exp Biol. 2016; 219: 1675–1688. https://doi.org/10.1242/jeb.135665 PMID: 27207644
- Deutsch CA, Tewksbury JJ, Huey RB, Sheldon KS, Ghalambor CK, Haak DC, et al. Impacts of climate warming on terrestrial ectotherms across latitude. Proc Natl Acad Sci. 2008; 105: 6668–6672. <u>https:// doi.org/10.1073/pnas.0709472105</u> PMID: <u>18458348</u>
- 78. Overgaard J, Kristensen TN, Mitchell KA, Hoffmann AA, Geber SEMA. Thermal tolerance in widespread and tropical Drosophila Species: does phenotypic plasticity increase with latitude? Am Nat. 2011; 178: S80–S96. https://doi.org/10.1086/661780 PMID: 21956094
- 79. Moller-Jacobs LL, Murdock CC, Thomas MB. Capacity of mosquitoes to transmit malaria depends on larval environment. Parasit Vectors. 2014; 7: 593. <u>https://doi.org/10.1186/s13071-014-0593-4</u> PMID: 25496502
- Thomas CD, Cameron A, Green RE, Bakkenes M, Beaumont LJ, Collingham YC, et al. Extinction risk from climate change. Nature. 2004; 427: 145–148. <u>https://doi.org/10.1038/nature02121</u> PMID: <u>14712274</u>
- Johansson F, Orizaola G, Nilsson-Örtman V. Temperate insects with narrow seasonal activity periods can be as vulnerable to climate change as tropical insect species. Sci Rep. 2020; 10: 8822. <u>https://doi.org/10.1038/s41598-020-65608-7</u> PMID: 32483233
- 82. Couper L, Farner J, Caldwell J, Childs M, Harris M, Kirk D, et al. How will mosquitoes adapt to climate change? Preprints; 2020 Nov. https://doi.org/10.22541/au.160589900.06282166/v1
- Sternberg ED, Thomas MB. Local adaptation to temperature and the implications for vector-borne diseases. Trends Parasitol. 2014; 30: 115–122. https://doi.org/10.1016/j.pt.2013.12.010 PMID: 24513566
- 84. Gray EM. Thermal acclimation in a complex life cycle: The effects of larval and adult thermal conditions on metabolic rate and heat resistance in *Culex pipiens* (Diptera: Culicidae). J Insect Physiol. 2013; 59: 1001–1007. <u>https://doi.org/10.1016/j.jinsphys.2013.08.001</u> PMID: 23932965
- Turelli M, Barton NH. Deploying dengue-suppressing *Wolbachia*: robust models predict slow but effective spatial spread in *Aedes aegypti*. Theor Popul Biol. 2017; 115: 45–60. <u>https://doi.org/10.1016/j.tpb.</u> 2017.03.003 PMID: 28411063
- 86. Caragata EP, Dutra HLC, Moreira LA. Exploiting intimate relationships: controlling mosquito transmitted disease with *Wolbachia*. Trends Parasitol. 2016; 32: 207–218. <u>https://doi.org/10.1016/j.pt.2015.10.011</u> PMID: <u>26776329</u>
- O'Neill SL, Ryan PA, Turley AP, Wilson G, Retzki K, Iturbe-Ormaetxe I, et al. Scaled deployment of Wolbachia to protect the community from dengue and other Aedes transmitted arboviruses. Gates Open Res. 2018; 2: 36. <u>https://doi.org/10.12688/gatesopenres.12844.3</u> PMID: <u>30596205</u>
- Frentiu FD, Robinson J, Young PR, McGraw EA, O'Neill SL. *Wolbachia*-mediated resistance to dengue virus infection and death at the cellular level. PLOS ONE. 2010; 5. <u>https://doi.org/10.1371/journal.pone.</u> 0013398 PMID: 20976219
- Clemson AS, Sgrò CM, Telonis-Scott M. Thermal plasticity in *Drosophila melanogaster* populations from eastern Australia: quantitative traits to transcripts. J Evol Biol. 2016; 29: 2447–2463. <u>https://doi.org/10.1111/jeb.12969</u> PMID: 27542565

90. Ford SA, Allen SL, Ohm JR, Sigle LT, Sebastian A, Albert I, et al. Selection on Aedes aegypti alters Wolbachia-mediated dengue virus blocking and fitness. Nat Microbiol. 2019; 4: 1832–1839. <u>https://doi.org/ 10.1038/s41564-019-0533-3</u> PMID: <u>31451771</u>